Author's response to reviews

Title: Ultrasound Study of Carotid and Cardiac Remodeling and Cardiac-arterial Coupling in Normal Pregnant and Preeclampsia Women

Authors:

Li-Jun Yuan (yuanlj@fmmu.edu.cn)
Yun-You Duan (duanyy@fmmu.edu.cn)
Dan Xue (ly8163703@163.com)
Tie-Sheng Cao (caots@fmmu.edu.cn)

Version: 3 Date: 29 January 2014

Author's response to reviews: see over
Dear Ms Janelyn Ann Cruz and dear Dr Robin Gandley,
Thank you very much for your email. I really appreciate your advice on revising the manuscript. We asked a sophisticated English language teacher to help us correct the language, especially the language of the abstract. Please check. We did not highlight the language changes this time. The following is the point-to-point responses to the questions by the reviewers. Thank you for all your great efforts on helping to improve our paper. If there is anything else that we need to do, please let us know. Thank you.

Yours sincerely,
Li-Jun Yuan

Reviewer Asma khali
Question: The authors addressed my comments. There are some limitations in the methodology, e.g. the measurements were performed in supine position, which are not possible to correct for at this stage. I would suggest that the authors mention these limitations in the Discussion section.
Response: Thank you for your reminding. This limitation has been added into the section of “Limitations”, on Page 13, which were highlighted with yellow color. Please check. Thank you.

Reviewer Catalina De Paco
This reviewer does not have questions this time.